# **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1029/2020GL089027

#### **Key Points:**

- Double tropopause characteristics and the connection to ENSO are analyzed using vertically high resolved GPS RO observations
- More double tropopauses are detected during the cold La Niña phase and less during El Niño
- During La Niña, a much lower first lapse rate tropopause corresponds to a narrowing of the tropical belt around the Niño 3 region

#### Correspondence to:

H. Wilhelmsen, hallgeir.wilhelmsen@uni-graz.at

#### **Citation**:

Wilhelmsen, H., Ladstädter, F., Schmidt, T., & Steiner, A. K. (2020). Double tropopauses and the tropical belt connected to ENSO. *Geophysical Research Letters*, 47, e2020GL089027. https://doi.org/10.1029/2020GL089027

Received 25 MAY 2020 Accepted 13 JUN 2020 Accepted article online 18 JUN 2020

©2020. The Authors. This is an open access article under the terms of the Creative Commons

Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Double Tropopauses and the Tropical Belt Connected to ENSO

Hallgeir Wilhelmsen<sup>1,2,3</sup>, Florian Ladstädter<sup>1,3</sup>, Torsten Schmidt<sup>4</sup>, and Andrea K. Steiner<sup>1,2,3</sup>

<sup>1</sup>Wegener Center for Climate and Global Change (WEGC), University of Graz, Graz, Austria, <sup>2</sup>FWF-DK Climate Change, University of Graz, Graz, Austria, <sup>3</sup>Institute for Geophysics, Astrophysics, and Meteorology/Institute of Physics (IGAM/ IP), Graz, Austria, <sup>4</sup>GeoForschungsZentrum Potsdam (GFZ), Potsdam, Germany

**Abstract** A detailed analysis of double tropopause (DT) occurrences requires vertically well resolved, accurate, and globally distributed information on the troposphere-stratosphere transition zone. Here, we use radio occultation observations from 2001 to 2018 with such properties. We establish a connection between El Niño-Southern Oscillation (ENSO) phases and the distribution of DTs by analyzing the global and seasonal DT characteristics. The seasonal distribution of DTs reveals several hotspot locations, such as near the subtropical jet stream and over high mountain ranges, where DTs occur particularly often. In this study, we detect a higher number of DTs during the cold La Niña state while warmer El Niño events result in lower DT rates, affecting the structure of the tropopause region. Close to the Niño 3 region, this relates to a much lower first lapse rate tropopause altitude during La Niña and corresponds to an apparent narrowing of the tropical belt there.

**Plain Language Summary** Double tropopauses (DTs) are associated with several phenomena in the transition zone between the troposphere and the stratosphere. DTs may occur when midlatitudinal and tropical air masses meet and warmer air overlaps colder air. A detailed analysis of DT occurrences requires accurate and globally distributed temperature information with high vertical resolution. In this study, we use radio occultation temperature observations from 2001 to 2018 with such properties. This enables us to establish a connection between the El Niño-Southern Oscillation (ENSO) phases and the distribution of DTs and to analyze the global and seasonal DT characteristics. The seasonal distribution of DTs reveals locations where DTs occur particularly often. During the colder La Niña state of the ENSO, a higher number of DTs is detected, while the warmer El Niño events result in lower DT rates. Close to the tropics in the Pacific Ocean, this relates to a much lower tropopause height during La Niña, which corresponds to an apparent narrowing of the tropical belt.

### **1. Introduction**

The tropopause, the transition zone between the troposphere and the stratosphere, is of substantial importance to the exchange between these two atmospheric regimes (Holton et al., 1995). Depending on season and latitude, the tropopause is typically found at around 16 km in the tropics and at around 9 km at high latitudes (e.g., Schmidt et al., 2005; Seidel & Randel, 2006). At midlatitudes, the higher tropical tropopause domain may overlap the lower high-latitude tropopause domain and form double tropopauses (DTs), either because the high-latitude tropopause domain extends equatorward (Peevey et al., 2014; Wang & Polvani, 2011) or because the tropical tropopause domain extends poleward (Homeyer et al., 2010; Castanheira et al., 2012; Liu & Barnes, 2018; Pan et al., 2009; Randel et al., 2007). The more complex structure and variability of the upper troposphere and lower stratosphere (UTLS) region at midlatitudes, related to this overlap and the existence of DTs there, is key to understand the stratosphere-troposphere exchange (e.g., Boothe & Homeyer, 2017).

DT events are found especially frequently at midlatitudes in both hemispheres, in storm track regions, on the poleward side of the subtropical jet stream (STJ), more frequently during winter (Bischoff et al., 2007; Schmidt et al., 2006; Seidel & Randel, 2006), and over land (Schmidt et al., 2006). The STJ is of special interest, not only because it marks a region where midlatitudinal and tropical air meet, but the STJs are linked to Rossby wave breaking events, which again are associated to DTs and stratosphere-troposphere exchange



(e.g., Castanheira & Gimeno, 2011; Peevey et al., 2012). The exchange is enhanced when DTs are present (Liu & Barnes, 2018).

DTs have also been found in association with mountain gravity waves (Schmidt et al., 2006), cyclogenesis (Añel et al., 2008) and the upward vertical motion in warm conveyor belts in strong cyclonic circulation systems (Peevey et al., 2014; Wang & Polvani, 2011). DTs are linked to the strength of the upward branch of the Brewer-Dobson circulation (Castanheira et al., 2012) and are detected in cloud-top inversion layers (Biondi et al., 2012).

Previous studies (e.g., Reid & Gage, 1985; Rieckh et al., 2014) have shown a strong correlation between the altitude of the first lapse rate tropopause (LRT) and the El Niño-Southern Oscillation (ENSO). Castanheira et al. (2012) showed a clear signal of both the quasi-biennial oscillation (QBO) and the ENSO in DT frequencies from reanalyzed data (ERA-Interim).

However, a systematic analysis of globally distributed DTs from observations and their link to ENSO phases has not yet been conducted. Previous studies on multiple tropopauses have mainly used radiosonde or reanalysis data. Due to the sparse sampling of radiosondes, a global view on the characteristics of DTs is hardly possible. Data with lower vertical resolution or smoother temperature profiles tend to smear out DT features and underestimate DT frequencies (Biondi et al., 2012; Manney et al., 2017; Vergados et al., 2014; Xian et al., 2019). Since the early 2000s, however, globally distributed measurements from radio occultation (RO) are available. This limb sounding technique provides temperature profiles with high accuracy and vertical resolution in the UTLS for applications in atmospheric research and climate (Anthes, 2011; Steiner et al., 2011). Some studies have used a subset of available RO data to study DTs (e.g. Lakkis & Canziani, 2009; Randel et al., 2007; Schmidt et al., 2006; Xu et al., 2014).

In this study, we take advantage of the precise DT detection with RO to present the global and seasonal characteristics of DTs from observations. For the first time, we analyze the relation between DTs and ENSO events and its implication on the tropopause structure based on the recent full multiyear RO record, which now covers 11 ENSO events. This larger sample substantially improves our ability to investigate these relations.

### 2. Data

Due to their high vertical resolution and global distribution, RO satellite measurements are well suited for investigating the thermal tropopause. RO measurements from different missions can be combined and used in continuation of each other (Foelsche et al., 2011; Schreiner et al., 2007). We used the Wegener Center OPSv5.6 data set (Angerer et al., 2017), a compilation of most RO satellite missions to date, enabling us to study DTs over a longer time period than previous studies. In this study, we use temperature profiles from September 2001 to December 2018, interpolated to an evenly spaced, fixed vertical grid with h = 100 m between the grid points.

El Niño and La Niña events are identified using the Oceanic Niño Index (ONI), which is a  $\pm 1$  running mean of monthly mean sea surface temperature anomalies in the Niño 3.4 region (5°S to 5°N and 170°W to 120°W). An event is called "El Niño" or "La Niña" when five succeeding months of the ONI are all above 0.5 K or all below -0.5 K, respectively.

In addition, we indicate the location of the STJ using the maximum mean horizontal wind speed between 200 and 300 hPa within 5° latitude  $\times$  5° longitude grid cells from European Centre for Medium-Range Weather Forecasts (ECMWF) 6-hourly analysis wind fields.

## 3. Methods

The standard definition of the World Meteorological Organization (WMO, 1957) was used to calculate the thermal LRTs: "(a) The first tropopause is defined as the lowest level at which the lapse rate decreases to 2°C/km or less, provided also the average lapse rate between this level and all higher levels within 2 km does not exceed 2°C/km. (b) If above the first tropopause the average lapse rate between any level and all higher levels within 1 km exceeds 3°C/km, then a second tropopause is defined by the same criterion as under (a). This tropopause may be either within or above the 1 km layer." This algorithm was applied to each RO



temperature profile on an evenly spaced 100-m grid. The search started from below, using a latitude,  $\varphi$ , dependent lower altitude limit,  $z_{\text{start}}$  (in meters), following

$$z_{\text{start}}(\varphi) = 6,250 \text{ m} + 1,250 \text{ m} \times \cos\varphi.$$
 (1)

This limit was adapted from Son et al. (2011) and adjusted downwards to include more LRTs at midlatitudes. The search was terminated at 25 km.

A candidate LRT altitude was first found on the evenly spaced 100-m vertical grid, according to the thresholds in the WMO (1957) definition. Due to the implementation, this point was always found at an altitude above the threshold. To avoid a positive bias, the altitude was fine tuned, by linear interpolation, down to the altitude where the lapse rate equals  $\Gamma = 2^{\circ}$ C km<sup>-1</sup>. This altitude was selected to represent the LRT.

The number of first LRTs,  $N_1$ , and the number of second LRTs,  $N_2$ , from all the available temperature profiles were counted and the corresponding DT percentages calculated using

DT percentage 
$$=\frac{N_2}{N_1} \times 100\%$$
. (2)

We calculated these percentages within  $5^{\circ} \times 5^{\circ}$  grid cells and within each  $5^{\circ}$  zonal band or each  $5^{\circ}$  meridional band. Finally, monthly mean DT anomalies were created relative to the mean seasonal cycle from 2007 to 2018.

#### 4. Results

#### 4.1. Global DT Occurrences

The main characteristics of the global DT distribution can be deduced from Figure 1. The figure reaffirms previous studies (e.g., Peevey et al., 2012, 2014; Randel et al., 2007; Schmidt et al., 2006; Wang & Polvani, 2011; Xu et al., 2014) and shows that the features are consistently revealed in the multiyear RO record. The seasonal features have been described in greater detail in Peevey et al. (2012). DTs are mainly found at midlatitudes along the STJ and are more frequent during winter and over land. The strong belt of DTs in the STJ regions gets weaker during summer, on both hemispheres, due to the weaker eddy activity as the summer STJ slows down. The STJs are primarily radiatively driven and move equatorward during winter (e.g., Maher et al., 2020; Manney et al., 2014), which is also the case for the location of the DT belt during December-January-February (DJF) (Figure 1a) and June-July-August (JJA) (Figure 1c).

The enhanced DT percentages in the tropical regions for all seasons may be explained by cloud tops (Biondi et al., 2012) or gravity waves in the stratosphere (e.g., Hoffmann et al., 2013 and references therein), both caused by deep convection in the tropics. Additionally, some of the DTs detected in the tropics are related to changing QBO phases (e.g., Kedzierski et al., 2016). We also detect these QBO-related DTs because we use a rather high upper altitude limit of 25 km when finding the LRTs, which is several kilometers above the mean LRT altitude, well into the QBO region in the stratosphere.

Furthermore, Figure 1 reveals locations where DTs are found particularly often. In the Northern Hemisphere (NH), such hotspots are located east of the Rocky Mountains, over the Himalayas, and over Japan. In the Southern Hemisphere (SH), they are found over and east of the southern Andes and over southeast Australia. All these DT hotspots are found on the STJ band, on the lee side of high mountains. They get weaker during summer, supporting the source to be mountain gravity waves related to the STJ (Schmidt et al., 2006). The enhancement east of Japan more or less disappears during JJA but is globally the strongest hotspot during DJF. The hotspot leeward of the Southern Andes is, as an exception, prominent for all seasons and is globally the strongest hotspot during JJA. The area is known for its high number of occurrences of gravity waves (Ern et al., 2018; Hoffmann et al., 2013; Sato et al., 2012).

To the *west* of the Andes, there is a DT *tail*, pointing toward the tropics, that only shows up during DJF (Figure 1a). As the main STJ flow is eastward, the tail is on the windward side of the Andes and therefore requires an alternative explanation. There may also be a similar, somewhat weaker, feature during JJA (Figure 1c) in the northeastern Pacific, west of the Rocky Mountains, also pointing toward the tropics.





Double tropopause percentages by season

**Figure 1.** Global spatial distribution of DTs from the years 2001 to 2018 by season: (a) December-January-February (DJF), (b) March-April-May (MAM), (c) June-July-August (JJA), and (d) September-October-November (SON). Cyan contour lines: Mean winds (of the maximum between 300 and 200 hPa) stronger than  $25 \text{ ms}^{-1}$  are indicated, with  $10 \text{ ms}^{-1}$  between the contours.



**Figure 2.** Monthly time series of DT percentages split into (a)  $5^{\circ}$  zonal bands, (b)  $5^{\circ}$  meridional bands, and (c) monthly mean anomalies of the meridional bands in panel (b). Only the DTs found between 60°S and 60°N were included in panels (b) and (c). Line in panel (c): The Oceanic Niño Index (ONI), with La Niña events in green and El Niño events in red. The right *y* axis in panel (c) corresponds to the ONI, in Kelvin.





Figure 3. Global spatial distribution of DTs during (a) DJF La Niña, (b) DJF El Niño, and (c) difference between DJF La Niña and DJF El Niño (i.e., a minus b).

#### 4.2. Annual Patterns of DTs

The seasonal DT development is exposed in Figure 2 in zonal (Figure 2a) and meridional (Figure 2b) bands. The noisy structure before 2006 is caused by the lower number of RO measurements in the early RO phase (see, e.g., Steiner et al., 2011), especially apparent in the meridional view in Figure 2b. The seasonality and clear zonal DT distribution mainly in the subtropics of both hemispheres become evident in Figure 2a. In both hemispheres, an annual pattern of peaks in DT numbers moves equatorward during winter, stopping at around  $\pm 30^{\circ}$  latitude and withdrawing poleward in summer. The DT rates in the NH are more pronounced, last longer, and cover a greater area than in the SH, due to the different land-sea distribution and thus different Rossby wave patterns (e.g., Schmidt et al., 2006; Xian et al., 2019).



# **Geophysical Research Letters**



**Figure 4.** Middle panels (a and b): The mean altitude of the first LRT for DJF only, during (a) La Niña and (b) El Niño. The blue rectangles mark the location of the Niño 3 region. Second topmost (i, j, k) and second bottommost panels (l, m, n): The location of all the first (blue) and the second (orange) LRTs found within the meridional bands marked by dashed lines in the middle panels. Red lines: The mean altitudes of the first LRT within these dashed lines. The red circles have the same location in each panel, for comparison. Topmost (c to h) and bottommost (o to t) panels: The temperature profiles from 25°S to 20°S and 20°N to 25°N (i.e., the profiles within the dashed lines in panels i to k and l to n).

The prominent straight line in Figure 2b, around 75°W, is attributed to the hotspot leeward of the southern Andes that shows up in every season in Figure 1. In contrast, the dip just west of 75°W reveals a meridional band with rarely any DTs, just to the west of the Andes, from the beginning to the end of the time series.



The irregularly recurring patterns further west in Figure 2b, between 150°W and 90°W, resemble ENSO time patterns. The peaks mainly show up between 30°S and 10°S and again between 10°N and 30°N, which is just north and south of the Niño 3 region (5°N to 5°S, 150°W to 90°W). Both the locations and the pattern suggest a link to the ENSO. In the following, we unravel more details and possible explanations.

#### 4.3. ENSO and DT Structure

The relation between ENSO and DT occurrences becomes evident in Figure 2c. First, it reveals that the warmer El Niño events result in lower DT rates. The indication of an ENSO connection is especially prominent for the El Niño events in 2009/2010 and 2015/2016. The ONI peak in 2006/2007 is not called an El Niño event because the values did not last long enough above the 0.5 K threshold, but nevertheless, the rate of DTs is also reduced. Second, the colder La Niña events lead to more DTs, especially between 150°W and 90°W.

For further investigation of the DT and ENSO relation, we limit the considered DTs to the DJF seasons only and examine El Niño and La Niña periods separately. This minimizes the seasonal influence on the difference between the two ENSO states. ENSO events occur more frequently during DJF, with 16 La Niña and 16 El Niño months detected in the considered time period. Figure 3 shows the global spatial distribution of DTs split into DJF La Niña events (Figure 3a) and DJF El Niño events (Figure 3b). It appears that the DT tail to the west of the Andes (cf. Figure 1a) mainly originates from the DJF La Niña time periods, as it is much weaker during DJF El Niño and the other time periods (not shown). Figure 3c shows the difference between the DT percentages during DJF La Niña and DJF El Niño, that is, the difference between Figures 3a and 3b. It uncovers that the tail is not only a SH feature but also shows up on the NH, disguised (in Figure 3a) by the high frequency of DTs in the latitudinal band around the NH STJ during DJF.

The increased DT occurrences in the eastern Pacific region during La Niña compared to El Niño periods are caused by distinct atmospheric circulation regimes. During La Niña, the upwelling part of the Walker circulation is located over the Maritime continent, while during El Niño the main upwelling moves to the central Pacific (see, e.g., Gettelman et al., 2001; Lau & Yang, 2015). An analysis of atmospheric parameters related to cyclonic activity (divergence and vorticity; not shown here) confirmed that during La Niña (El Niño), cyclonic (anticyclonic) activity is dominant in the tropopause region above the eastern Pacific. According to Randel et al. (2007), upper tropospheric level cyclonic vorticity is related to an enhancement of DT occurrence and lower LRT heights compared to anticyclonic vorticity at the same altitude level. This is in good agreement with the observed differences in Figure 3c.

Figure 3 additionally suggests that during DJF La Niña, more DTs should be expected at locations where they are usually found than during DJF El Niño. The regions just west of the Andes hotspot and east of the Rocky Mountains hotspot especially stand out. The enhancement east of Japan, on the other hand, seems to be unaffected by ENSO events. There is a southward shift around the Himalayan hotspot during DJF El Niño, also seen between the seasons (Figure 1), related to jet stream shifts (Maher et al., 2020).

#### 4.4. Narrowing the Tropical Belt During La Niña

We investigate the impact of ENSO conditions on the tropopause structure in Figure 4. Figures 4a and 4b show the mean of all the first LRT altitudes during DJF La Niña and DJF El Niño, respectively. For most longitudes, the mean first LRTs in the tropics are found at a slightly lower altitude during DJF La Niña (Figure 4a) than during DJF El Niño (Figure 4b). This is in agreement with Randel et al. (2007) (see above). Poleward of the Niño 3 region, they are exceptionally low, corresponding to a remarkable narrowing of the tropical belt around those longitudes. Figure 4i shows the altitude of all the first (blue) and second (orange) LRTs within a 10° meridional band at the Niño 3 region (leftmost dashed meridional band in Figure 4a), at their corresponding latitudes during La Niña. Additional tropopauses appear below the tropical tropopause domain (approximately 20°S to 20°N) from what seems to be an equatorward expansion of the high-latitude tropopause domain. The tropical LRT domain is still present, as a mix of first and second LRTs, but the mean first LRT appears much lower than usual in the specified region for DJF La Niña. The corresponding temperature profile cluster plots in Figures 4c-4h (DJF La Niña) and Figures 4o-4t (DJF El Niño) support these observations. This becomes especially clear in Figure 4d, where the temperature profiles start off with a steady lapse rate until they are sliced in two "branches" at the first LRT and break into typical, although less sharply defined, tropical temperature profiles (cf. Figure 4g). For comparison, the regular tropopause characteristics are depicted for two additional meridional bands in Figure 4.



Similar to the DT frequency as described in the previous section, the altitude of the first LRT is also influenced by the circulation patterns. As cyclonic vorticity leads to lower first LRT altitudes (Randel et al., 2007), the shifting cyclonic circulation patterns during La Niña, induced by the change in the Walker circulation, are therefore considered to be connected to the observed tropical belt narrowing.

#### 5. Conclusions

To summarize, high vertical resolution and global coverage make RO satellite measurements well suited for studying DTs. The measurements are especially accurate in the altitude range of the tropopause. We exploited these characteristics to investigate for the first time the relation of ENSO and the occurrence of DTs in global multiyear RO observations.

We demonstrated that findings from previous studies are consistently revealed and presented a seasonally and regionally resolved picture of various DT hotspots. DTs are mainly triggered along the STJ in winter, especially prominent over the Rocky Mountains, the Himalayas, and the Andes.

Temporal, resolved DT patterns introduced a connection between ENSO and DT occurrences. It revealed an increase in DTs to the west of the Andes, poleward of the Niño 3 region (5°N to 5°S, 150°W to 90°W). We found that the strength and location of this increase is evidently connected to the ENSO. Colder La Niña events lead to a higher number of DTs while warmer El Niño events result in lower DT rates. This difference in DT occurrences is considered to be caused by the changing atmospheric circulation regimes of the Walker circulation between the ENSO phases.

During La Niña, the higher number of DTs detected at the Niño 3 region seems to be an equatorward expansion of the high-latitude tropopause domain. This leads to a mix of first and second LRTs at the edge of the tropical belt and a mean first LRT altitude that is much lower than usual. This corresponds to an apparent narrowing of the tropical belt there.

Knowledge of the detailed structure of the UTLS region is of great relevance for the analysis of the stratosphere-troposphere exchange. Identifying regions of increased DT occurrences points to a possible enhanced exchange. This has implications for the composition of the atmosphere, influencing, for example, the radiative balance and the dynamics of the atmosphere (see, e.g., Stohl et al., 2003 and references therein). The enhanced DT frequencies and lower first LRTs suggest that these processes are of increased relevance in the tropical Eastern Pacific during La Niña.

Recent studies have discussed the widening of the tropical belt (e.g., Staten et al., 2018 and references therein). This widening is difficult to determine due to large internal variability. The dependence of the tropical belt width on ENSO presented in this study might be of relevance to future studies on this topic.

Compared to neutral ENSO phases, ENSO events substantially alter the UTLS structure at midlatitudes and the tropics. For a detailed analysis, vertically high resolved information with global coverage is needed. Our results show that RO observations are able to provide such analysis and contribute to gaining improved knowledge of the transition between troposphere and stratosphere and the variability of the tropical belt.

#### References

- Añel, J., Antuña, J. C., de la Torre, L., Castanheira, J. M., & Gimeno, L. (2008). Climatological features of global multiple tropopause events. Journal of Geophysical Research, 113, D00B08. https://doi.org/10.1029/2007JD009697
- Angerer, B., Ladstädter, F., Scherllin-Pirscher, B., Schwärz, M., Steiner, A. K., Foelsche, U., & Kirchengast, G. (2017). Quality aspects of the WEGC multi-satellite GPS radio occultation record OPSv5.6. Atmospheric Measurement Techniques, 10, 4845–4863. https://doi.org/ 10.5194/amt-10-4845-2017
- Anthes, R. A. (2011). Exploring Earth's atmosphere with radio occultation: contributions to weather, climate, and space weather. *Atmospheric Measurement Techniques*, 4, 1077–1103. https://doi.org/10.5194/amt-4-1077-2011
- Biondi, R., Randel, W. J., Ho, S. P., Neubert, T., & Syndergaard, S. (2012). Thermal structure of intense convective clouds derived from GPS radio occultations. Atmospheric Chemistry and Physics, 12(12), 5309–5318. https://doi.org/10.5194/acp-12-5309-2012
- Bischoff, S. A., Canziani, P. O., & Yuchechen, A. E. (2007). The tropopause at southern extratropical latitudes: Argentine operational rawinsonde climatology. *International Journal of Climatology*, *27*(2), 189–209. https://doi.org/10.1002/joc.1385
- Boothe, A. C., & Homeyer, C. R. (2017). Global large-scale stratosphere-troposphere exchange in modern reanalyses. *Atmospheric Chemistry and Physics*, 17(9), 5537–5559. https://doi.org/10.5194/acp-17-5537-2017
- Castanheira, J. M., & Gimeno, L. (2011). Association of double tropopause events with baroclinic waves. *Journal of Geophysical Research*, 116, D19113. https://doi.org/10.1029/2011JD016163

#### Acknowledgments

The authors express thanks to the WEGC EOPAC team for providing the OPSv5.6 RO data set (https://doi.org/ 10.25364/WEGC/OPS5.6:2019.1), to NOAA for the Oceanic Niño Index (https://www.cpc.ncep.noaa.gov/data/ indices/oni.ascii.txt), to ECMWF for wind fields (https://www.ecmwf.int/ en/forecasts/datasets), and to Thomas Birner for pointing to a more robust interpretation of the LRT definition than initially used. This work was funded by the Austrian Science Fund (FWF) under Research Grants P27724-NBL (VERTICLIM) and W1256-G15 (Doctoral Programme Climate Change-Uncertainties, Thresholds and Coping Strategies). Finally, a special thanks to the open source community, particularly those funded by NumFOCUS.



- Castanheira, J. M., Peevey, T. R., Marques, C. A. F., & Olsen, M. A. (2012). Relationships between Brewer-Dobson circulation, double tropopauses, ozone and stratospheric water vapour. *Atmospheric Chemistry and Physics*, 12, 10,195–10,208. https://doi.org/10.5194/ acp-12-10195-2012
- Ern, M., Trinh, Q. T., Preusse, P., Gille, J. C., Mlynczak, M. G., Russell III, J. M., & Riese, M. (2018). GRACILE: A comprehensive climatology of atmospheric gravity wave parameters based on satellite limb soundings. *Earth System Science Data*, 10(2), 857–892. https://doi.org/10.5194/essd-10-857-2018
- Foelsche, U., Scherllin-Pirscher, B., Ladstädter, F., Steiner, A. K., & Kirchengast, G. (2011). Refractivity and temperature climate records from multiple radio occultation satellites consistent within 0.05 %. Atmospheric Measurement Techniques, 4, 2007–2018. https://doi.org/10.5194/amt-4-2007-2011
- Gettelman, A., Randel, W. J., Massie, S., Wu, F., Read, W. G., & Russell III, J. M. (2001). El Niño as a natural experiment for studying the tropical tropopause region. *Journal of Climate*, 14, 3375–3392.
- Hoffmann, L., Xue, X., & Alexander, M. J. (2013). A global view of stratospheric gravity wave hotspots located with Atmospheric Infrared Sounder observations. Journal of Geophysical Research: Atmosphere, 118, 416–434. https://doi.org/10.1029/2012JD018658
- Holton, J. R., Haynes, P. H., McIntyre, M. E., Douglass, A. R., Rood, R. B., & Pfister, L. (1995). Stratosphere-troposphere exchange. *Reviews of Geophysics*, 33(14), 403–439. https://doi.org/10.1029/95RG02097
- Homeyer, C. R., Bowman, K. P., & Pan, L. L. (2010). Extratropical tropopause transition layer characteristics from high-resolution sounding data. Journal of Geophysical Research, 115, D13108. https://doi.org/10.1029/2009JD013664
- Kedzierski, R. P., Matthes, K., & Bumke, K. (2016). The tropical tropopause inversion layer: Variability and modulation by equatorial waves. Atmospheric Chemistry and Physics, 16(18), 11,617–11,633. https://doi.org/10.5194/acp-16-11617-2016
- Lakkis, S., & Canziani, P. (2009). A comparative analysis of the temperature behavior and multiple tropopause events derived from GPS, radiosonde and reanalysis datasets over Argentina, as an example of Southern mid latitudes. *Revista de Climatología*, 9, 1–14.
- Lau, K. M., & Yang, S. (2015). Tropical meteorology & climate | Walker circulation, (2nd ed.). In G. R. North, J. Pyle, & F. Zhang (Eds.), Encyclopedia of atmospheric sciences (pp. 177–181). Oxford: Academic Press. https://doi.org/10.1016/B978-0-12-382225-3.00450-3
- Liu, C., & Barnes, E. (2018). Synoptic formation of double tropopauses. Journal of Geophysical Research: Atmosphere, 123, 693–707. https:// doi.org/10.1002/2017JD027941
- Maher, P., Kelleher, M. E., Sansom, P. G., & Methven, J. (2020). Is the subtropical jet shifting poleward? *Climate Dynamics*, 54(3), 1741–1759. https://doi.org/10.1007/s00382-019-05084-6
- Manney, G. L., Hegglin, M. I., Daffer, W. H., Schwartz, M. J., Santee, M. L., & Pawson, S. (2014). Climatology of upper tropospheric-lower stratospheric (UTLS) jets and tropopauses in MERRA. *Journal of Climate*, 27, 3248–3271. https://doi.org/10.1175/JCLI-D-13-00243.1
- Manney, G. L., Hegglin, M. I., Lawrence, Z. D., Wargan, K., Millán, L. F., Schwartz, M. J., & Daffer, W. H. (2017). Reanalysis comparisons of upper tropospheric-lower stratospheric jets and multiple tropopauses. *Atmospheric Chemistry and Physics*, 17(18), 11,541–11,566. https://doi.org/10.5194/acp-17-11541-2017
- Pan, L., L., Randel, W. J., Gille, J. C., Hall, W. D., Nardi, B., et al. (2009). Tropospheric intrusions associated with the secondary tropopause. Journal of Geophysical Research, 114, D10302. https://doi.org/10.1029/2008JD011374
- Peevey, T. R., Gille, J. C., Homeyer, C. R., & Manney, G. L. (2014). The double tropopause and its dynamical relationship to the tropopause inversion layer in storm track regions. *Journal of Geophysical Research: Atmosphere*, 119, 10,194–10,212. https://doi.org/10.1002/ 2014JD021808
- Peevey, T. R., Gille, J. C., Randall, C. E., & Kunz, A. (2012). Investigation of double tropopause spatial and temporal global variability utilizing High Resolution Dynamics Limb Sounder temperature observations. *Journal of Geophysical Research*, 117, D01105. https://doi. org/10.1029/2011JD016443
- Randel, W. J., Seidel, D. J., & Pan, L. L. (2007). Observational characteristics of double tropopauses. Journal of Geophysical Research, 112, D07309. https://doi.org/10.1029/2006JD007904
- Reid, G. C., & Gage, K. S. (1985). Interannual variations in the height of the tropical tropopause. Journal of Geophysical Research, 90(D3), 5629–5635. https://doi.org/10.1029/JD090iD03p05629
- Rieckh, T., Scherllin-Pirscher, B., Ladstädter, F., & Foelsche, U. (2014). Characteristics of tropopause parameters as observed with GPS radio occultation. Atmospheric Measurement Techniques, 7, 3947–3958. https://doi.org/10.5194/amt-7-3947-2014
- Sato, K., Tateno, S., Watanabe, S., & Kawatani, Y. (2012). Gravity wave characteristics in the Southern Hemisphere revealed by a high-resolution middle-atmosphere general circulation model. *Journal of the Atmospheric Sciences*, 69(4), 1378–1396. https://doi.org/ 10.1175/JAS-D-11-0101.1
- Schmidt, T., Beyerle, G., Heise, S., Wickert, J., & Rothacher, M. (2006). A climatology of multiple tropopauses derived from GPS radio occultations with CHAMP and SAC-C. *Geophysical Research Letters*, 33, L04808. https://doi.org/10.1029/2005GL024600
- Schmidt, T., Heise, S., Wickert, J., Beyerle, G., & Reigber, C. (2005). GPS radio occultation with CHAMP and SAC-C: Global monitoring of thermal tropopause parameters. Atmospheric Chemistry and Physics, 5(6), 1473–1488. https://doi.org/10.5194/acp-5-1473-2005
- Schreiner, W., Rocken, C., Sokolovskiy, S., Syndergaard, S., & Hunt, D. (2007). Estimates of the precision of GPS radio occultations from the COSMIC/FORMOSAT-3 mission. *Geophysical Research Letters*, *34*, L04808, https://doi.org/10.1029/2006GL027557
- Seidel, D. J., & Randel, W. J. (2006). Variability and trends in the global tropopause estimated from radiosonde data. *Journal of Geophysical Research*, 111, D21101. https://doi.org/10.1029/2006JD007363
- Son, S. W., Tandon, N. F., & Polvani, L. M. (2011). The fine-scale structure of the global tropopause derived from COSMIC GPS radio occultation measurements. *Journal of Geophysical Research*, *116*, D20113. https://doi.org/10.1029/2011JD016030
- Staten, P. W., Lu, J., Grise, K. M., Davis, S. M., & Birner, T. (2018). Re-examining tropical expansion. Nature Climate Change, 8(43), 768–775. https://doi.org/10.1038/s41558-018-0246-2
- Steiner, A. K., Lackner, B. C., Ladstädter, F., Scherllin-Pirscher, B., Foelsche, U., & Kirchengast, G. (2011). GPS radio occultation for climate monitoring and change detection. *Radio Science*, 46, RS0D24. https://doi.org/10.1029/2010RS004614
- Stohl, A., Bonasoni, P., Cristofanelli, P., Collins, W., Feichter, J., Frank, A., & Zerefos, C. (2003). Stratosphere-troposphere exchange: A review, and what we have learned from STACCATO. *Journal of Geophysical Research*, 108(D12), 8516. https://doi.org/10.1029/ 2002JD002490
- Vergados, P., Luo, Z. J., Emanuel, K., & Mannucci, A. J. (2014). Observational tests of hurricane intensity estimations using GPS radio occultations. Journal of Geophysical Research: Atmosphere, 119, 1936–1948. https://doi.org/10.1002/2013JD020934
- WMO (1957). Meteorology—A three dimensional science: Second session of the commission for aerology. Geneva: World Meteorological Organization (WMO). (WMO Bull. No. 4).
- Wang, S., & Polvani, L. M. (2011). Double tropopause formation in idealized baroclinic life cycles: The key role of an initial tropopause inversion layer. *Journal of Geophysical Research*, 116, D05108. https://doi.org/10.1029/2010JD015118

Xian, T., & Homeyer, C. R. (2019). Global tropopause altitudes in radiosondes and reanalyses. Atmospheric Chemistry and Physics, 19(8), 5661–5678. https://doi.org/10.5194/acp-19-5661-2019

Xu, X., Gao, P., & Zhang, X. (2014). Global multiple tropopause features derived from COSMIC radio occultation data during 2007 to 2012. Journal of Geophysical Research: Atmosphere, 119, 8515–8534. https://doi.org/10.1002/2014JD021620